Author's response to reviews

**Title:** Functional characterisation of the TSC1-TSC2 complex to assess multiple TSC2 variants identified in single families affected by tuberous sclerosis complex

**Authors:**

Mark Nellist (m.nellist@erasmusmc.nl)
Ozgur Sancak (Ozgur.Sancak@dsm.com)
Miriam Goedbloed (m.goedbloed@erasmusmc.nl)
Alwin Afdriaans (a.adriaans@erasmusmc.nl)
Marja Wessels (m.w.wessels@erasmusmc.nl)
Anneke Maat-Kievit (j.a.maat@erasmusmc.nl)
Marieke Baars (m.j.baars@amc.uva.nl)
Charlotte Dommering (c.j.dommering@vumc.nl)
Ans van den Ouweland (a.vandenouweland@erasmusmc.nl)
Dicky Halley (d.halley@erasmusmc.nl)

**Version:** 2  **Date:** 16 October 2007

**Author's response to reviews:**

Dear Sir/Madam,

Abstract has been revised into sections with the headings 'Background', 'Results' and 'Conclusions', as requested.

Yours sincerely,

Mark Nellist